Author's response to reviews

Title: The in vitro antisickling and antioxidant effects of aqueous extracts Zanthoxyllum heitzii on sickle cell disorder

Authors:

Nanfack P Nanfack NP (pauline600@yahoo.fr)
Prosper N Biapa BNPC (brbiapa@yahoo.fr)
Constant A Pieme PCA (apieme@yahoo.fr)
Jocelyne V Ama AMVJ (movicky@yahoo.fr)
Bruno M Moukette MMB (mouket2006@yahoo.fr)
Jeanne Y Ngogang NYJ (ingogang@yahoo.fr)

Version: 2 Date: 12 June 2013

Author's response to reviews: see over
Cover letter with reviewers and editor`s comments
By PIEME constant Anatole

1- Reviewer: Charles wambebe

I had reviewed this manuscript before. The authors responded to my suggestions. I have made some minor corrections in the revised manuscript which are indicated in blue color

Answer: Minors corrections indicated by this reviewer have been added in the text

2- Editors comments

The manuscript has been corrected regarding the guideline of the journal

1- The filiation of the authors has been corrected
2- The references have been edited
3- The legends of figures have been posted at the end of the manuscript